Author's response to reviews

Title: Trends in HIV infection surveillance among men who have sex with men in Norway, 1995-2009

Authors:

Irena Jakopanec (irena.jakopanec@dnv.com)
Andrej M Grijbovski (angr@fhi.no)
Oivind Nilsen (oivind.nilsen@fhi.no)
Hans Blystad (hans.blystad@fhi.no)
Preben Aavitsland (preben@epidemi.no)

Version: 2 Date: 16 July 2012

Author's response to reviews: see over
Dear Editors,

Upon your request, we have now added details on how we have accessed the Norwegian HIV surveillance data in the Methods section (Methods, paragraph 1) and how other researchers with similar interests could access these data. We hope we provided sufficient clarification, so that you can now initiate peer review process.

We look forward to hearing from you,

Dr. Andrej M. Grjibovski